Author's response to reviews

Title: The effectiveness of pelvic floor muscle training alone and with other physical therapies for the treatment of stress urinary incontinence in women

Authors:

Patricia B Neumann (cpneumann@ozemail.com.au)
Professor Dr Karen A Grimmer (Karen.grimmer@unisa.edu.au)
Yamini Deenadayalan (Yamini.deenadayalan@unisa.edu.au)

Version: 2 Date: 10 October 2005

Author's response to reviews:

Dr Sandra Le Centre for Allied Health Evidence
Biomedcentral - Women's Health University of South Australia

Cover Letter
Dear Dr Le,

Re Manuscript No: 7043624528121851
Thank you for this opportunity to resubmit our manuscript to BMC Women's Health. The document has been reformatted in portrait view as requested. A number of other minor corrections were also undertaken as detailed below:

- Whole document double spaced
- Title page: correction of Centre of Allied Health Evidence
- Abstract: page 4, line 8: (rephrased) The contribution of adjunctive therapies is unclear and there is limited evidence about treatment outcomes in primary care settings.
- Page 3, line 19: (word order) the eligibility of each study, its level of evidence and the methodological quality.
- Page 42, line 19: additional reference No 61
- Page53, line 46: reference 61 added
- Formatting changes to the tables, change font size
- Formatting changes to appendices from Ariel to Times New Roman and to fit portrait view.
- Missing row of data added to Appendix 4
- Title of Figure 1 shortened to 15 words and key added.
- Reformattng all tables to Portrait from Landscape
- Font size of the contents in all tables were changed from 12 to 10

We trust that these changes meet with your approval. Please let us know if there are any further problems. Yours sincerely,

Patricia Neumann
On behalf of Prof Karen Grimmer